Author’s response to reviews

Title: Epigenetic loss of heterozygosity of Apc and an inflammation-associated mutational signature detected in Lrig1+/−-driven murine colonic adenomas

Authors:

Jessica Preston (jpreston@molbio.uoregon.edu;jpreston555@gmail.com;jpreston@uoregon.edu)
Nicholas Stiffler (nstiffle@uoregon.edu)

Version: 3 Date: 04 Feb 2020

Author’s response to reviews:

Editor Comments: Please include an animal ethics statement in the 'Ethics approval and consent to participate' section of the Declarations.

The following ethics statement is now included in the ‘Ethics approval and consent to participate’ section: “All mouse experiments were approved by Institutional Animal Care and Use Committee (IACUC) at Vanderbilt University Medical Center. Animal studies were performed according to the recommendations in the Guide for the Care and Use of Laboratory Animals of the National Institutes of Health.”